

Revista de Psiquiatría y Salud Mental

www.elsevier.es/saludmental



EDITORIAL

How to improve the integrity of clinical trial articles Cómo mejorar la integridad de los artículos de los ensayos clínicos

Rafael Dal-Ré

Unidad de Epidemiología, Instituto de Investigación Sanitaria-Hospital Universitario Fundación Jiménez Díaz, Universidad Autónoma de Madrid, Madrid, Spain

Scandalous news is sometimes published about the integrity of medical research. Thus in 2017 The New England Journal of Medicine published a letter stating that in a clinical trial promoted by the United States National Institutes of Health (NIH), in 6 countries and published in the same journal in 2014, it was proven that 30% of the patients recruited in Russia had not received the drug being studied.1 This fact led some people to ask themselves whether the samples of the drug in question had been sold in the country's black market, as something must have happened to them.² Although news like this is very striking, the bestknown problem regarding the (lack of) integrity in clinical trials arises in how their results are communicated. This is more important than particular details of how trials are conducted, as usually only the promoters or main researchers are aware of these aspects. The problem in question here is the "distorted publication" or "selective reporting" of the results of clinical trials³: reporting the most interesting results (usually the "positive" ones that have attained statistical significance), while not publishing some of the results that, although they were said to be relevant in the protocol, had not attained such significance. This is especially important in mental health clinical trials, given that they have certain well-known particularities. These include the use of psychometric scales (which are therefore subject

To publish results in the more than 400 journals⁵ that belong to or have accepted the rules of the International Committee of Medical Journal Editors, clinical trials have to be registered in an open-access database (or registry) before the first participant is recruited. The registry with the largest number of clinical trials is the NIH clinicaltrials.gov, which contained more than 210,000 trials in January 2018. Registering trials prospectively is the best way of ensuring that results are published in an unbiased way. 6 Comparison of the variables and analysis published in papers with the ones included by researchers in the registry makes it possible to know if there are discrepancies (omissions, additions or changes) between what they said they were going to evaluate and what they eventually said they had evaluated. This therefore makes it possible to know whether a paper contains a distortion in its publication of results. Some recently published studies show a discouraging state of affairs. Thus only 14% of the trials published by the five most important psychiatry journals were registered prospectively and did not selectively communicate their results.⁷ 85% of the clinical trials which evaluated antipsychotic drugs showed discrepancies between what they published and what had been registered.8 The situation is even worse in psychotherapy, as 95% of the trials published in the five highest-impact clinical psychology journals had publication distortion. Lastly, 63% of papers made no mention at all of severe side effects, and when they did so, 10 49% had discrepancies between the papers and registrations. 11 Regarding the mortality rate during clinical trials of antidepressant and antipsychotic drugs, cases of death and suicide were only

E-mail address: Rafael.dalre@quironsalud.es

to a certain degree of subjectivity) as the main variables for measuring efficacy.⁴

^{*} Please cite this article as: Dal-Ré R. Cómo mejorar la integridad de los artículos de los ensayos clínicos. Rev Psiquiatr Salud Ment (Barc). 2018;11:189–191.

190 R. Dal-Ré

reported in a minority of the papers included in registries (38% and 47%, respectively).¹¹

However, do these discrepancies have any clinical relevance in terms of care? This is what Becker et al. ¹² found in 6% of 96 papers published in journals with an impact factor of at least 10. 55% of these papers were in fact published in *The New England Journal of Medicine*, *The Lancet* or the *JAMA*), and discrepancies in the main variable of efficacy were so great that they altered the interpretation of results. Although this percentage is striking, it could be expected to be even higher in papers published in journals with a less strict editorial policy.

Readers of medical journals have the right to expect that the information published will be true, exact and complete. To this end journals have to apply quality controls to prevent (or minimise) selective communication of results. Measures which are based on the responsibility of authors are not enough. 13 Quality control should consist of individuals independent of the authors, such as reviewers or the editorial team, checking the variables and analyses in manuscripts against those shown in the registry for each trial, or in the trial protocol, which is usually not available. 14 There are two problems here, as the majority of reviewers understand that this is not something they have to do, 15 and moreover the majority of journals do not even ask them to do it. 16 It may therefore be concluded that medical journals do not consider the selective reporting of the results of clinical trials to be a serious problem that they should correct. 17

Resolving the selective reporting distortion will have to be based on the commitment of everyone involved: promoters and financers, the hospitals where trials take place, researchers/authors and journals. There can be no doubt that over the long term the best solution is to train all of the healthcare professionals involved in clinical trials in the values of scientific integrity. 18 However, that would take a long time to produce results. Over the short to medium term the best way to prevent the proliferation of papers containing publication distortion would be to implement a preventive measure: to check, during the editorial process, that what the authors say in their paper corresponds to the information in the registry. It has to be taken into account that discrepancies which may arise may be scientifically necessary; in this case the authors have to state this and give reasons for it in their paper. ¹⁹ Given that editorial teams are unable to do this work, it has been suggested that mixed committees of clinical research experts and students do it, similar to the team that undertook the COMPare²⁰ project in the United Kingdom on the same subject. It has been calculated that from 150 to 200 teams would be necessary to do this checking work. They would be specialised in different therapeutic areas and belong to universities, research centres and scientific associations throughout the world. 19 These teams would work in the same way as external reviewers do now, interacting directly with the editorial teams of journals. The results of checking papers against their registries would be published as a supplement to the paper, so that readers would know if there were any discrepancies and, if there were any, the reasons for this¹⁹ which have sometimes been required during the editorial process.²¹

It is possible that many readers will consider that this proposal would complicate editorial reviewing by involving another agent to the process. Nevertheless, it is hard to imagine another solution over the short- to medium term when the credibility of clinical trials is questioned together with the correct interpretation of their results by clinicians and patients. On the other hand, similar measures have been applied in other areas of knowledge due to similar situations. For example, it is known that 4% of scientific papers contain duplicated images.²² Because of this The Journal of Biological Chemistry, which belongs to the American Society for Biochemistry and Molecular Biology, which is more than one hundred years old—and following the steps taken by The Journal of Cell Biology-, decided in 2017 to implement a preventive measure as a quality control: it contracted three people for its editorial team who, among other functions, examine the images contained in papers to detect any alteration or manipulation.²³ A tool has in fact already been proposed to aid this work.²⁴ If some journals have implemented measures like this to prevent the publication of duplicated images—a problem that affects 4% of images and which is of questionable and even minor relevance for public health—, what should medical journals be expected to do to prevent the publication of clinical trials that, in at least 6% of cases, may alter how their results are interpreted and therefore influence clinical decisions?

References

- De Denus S, O'Meara E, Desai AS, Claggett B, Lewis EF, Leclair G, et al. Spironolactone metabolites in TOPCAT — new insights into regional variation. N Engl J Med. 2017;376:1690-2.
- Lowe D. A clinical trial torpedoed by fraud and incompetence. In the pipeline. Science Transl Med blog; 27 April 2017. Available from: http://blogs.sciencemag.org/pipeline/archives/2017/04/27/a-clinical-trial-torpedoed-by-fraud-and-incompetence [access3rd edn1.01.18].
- Dal-Ré R, Marušić A. Prevention of selective outcome reporting: let us start from the beginning. Eur J Clin Pharmacol. 2016;72:1283-8.
- Dal-Ré R, Bobes J, Cuijpers P. Why prudence is needed when interpreting articles reporting clinical trial results in mental health. Trials. 2017;18:143.
- Hooft L, Korevaar DA, Molenaar N, Bossuyt PM, Scholten RJ. Endorsement of ICMJE's clinical trial registration policy: a survey among journal editors. Neth J Med. 2014;72:349–55.
- 6. Weber WE, Merino JG, Loder E. Trial registration 10 years on. BMJ. 2015;351, h3572.
- Scott A, Rucklidge JJ, Mulder RT. Is mandatory prospective trial registration working to prevent publication of unregistered trials and selective outcome reporting? An observational study of five psychiatry journals that mandate prospective clinical trial registration. PLoS ONE. 2015;10, e0133718.
- Lancee M, Lemmens CMC, Kahn RS, Vinkers CH, Luykx JJ. Outcome reporting bias in randomized-controlled trials investigating antipsychotic drugs. Transl Psychiatry. 2017;7, e1232.
- Bradley HA, Rucklidge JJ, Mulder RT. A systematic review of trial registration and selective outcome reporting in psychotherapy randomized controlled trials. Acta Psychiatr Scand. 2017;135:66-75.
- 10. De Vries YA, Roest AM, Beijers L, Turner EH, de Jonge P. Bias in the reporting of harms in clinical trials of second-generation antidepressants for depression and anxiety: a meta-analysis. Eur Neuropsychopharmacol. 2016;26:1752–9.
- Hughes S, Cohen D, Jaggi R. Differences in reporting serious adverse events in industry sponsored clinical trial registries and journal articles on antidepressant and antipsychotic drugs: a cross-sectional study. BMJ Open. 2014;4, e005535.

- Becker JE, Krumholz HM, Ben-Josef G, Ross JS. Reporting of results in ClinicalTrials.gov and high-impact journals. JAMA. 2014;311:1063-5.
- 13. Catalá-López F, Hutton B, Page MJ, Vieta E, Tabarés-Seisdedos R, Moher D. Declaración de transparencia: un paso hacia la presentación completa de artículos de investigación. Rev Psiquiatr Salud Ment (Barc). 2016;9:63–4.
- 14. Odutayo A, Altman DG, Hopewell S, Shakir M, Hsiao AJ, Emdin CA. Reporting of a publicly accessible protocol and its association with positive study findings in cardiovascular trials (from the Epidemiological Study Of Randomized Trials [ESORT]). Am J Cardiol. 2015;116:1280–3.
- **15.** Mathieu S, Chan AW, Ravaud P. Use of trial register information during the peer review process. PLoS ONE. 2013;356, e59910.
- 16. Chauvin A, Ravaud P, Baron G, Barnes C, Boutron I. The most important tasks for peer reviewers evaluating a randomized controlled trial are not congruent with the tasks most often requested by journal editors. BMC Med. 2015; 356:158.
- 17. Dal-Ré R, Caplan AL. Journal editors impasse with outcome reporting bias. Eur J Clin Invest. 2015;45:895–8.
- 18. ALLEA All European Academies. The European code of conduct for research integrity. Revised edition. Berlin;

- 2017. Available from: https://ec.europa.eu/research/participants/data/ref/h2020/other/hi/h2020-ethics_code-of-conduct_en.pdf [accessed 31.01. 18].
- 19. Ioannidis JP, Caplan AL, Dal-Ré R. Outcome reporting bias in clinical trials: why monitoring matters. BMJ. 2017;356, j408.
- 20. Goldacre B, Drysdale H, Dale A, Hartley P, Milosevic I, Slade E, et al. The COMPare project. Tracking switched outcomes in clinical trials. Available from: http://compare-trials.org/[accessed 31.01.18].
- Hopewell S, Witt CM, Linde K, Icke K, Adedire O, Kirtley S, et al. Influence of peer review on the reporting of primary outcome(s) and statistical analyses of randomised trials. Trials. 2018;19:30.
- 22. Bik EM, Casadevall A, Fang FC. The prevalence of inappropriate image duplication in biomedical research publications. MBio. 2016;7, pii:e00809-16.
- 23. McCook A. Job alert: biology society hiring editors to screen images. Retraction Watch; 21 April 2017. Available from: http://retractionwatch.com/2017/04/21/job-alert-biology-society-hiring-editors-screen-images/ [accessed 31.01.18].
- 24. Koppers L, Wormer H, Ickstadt K. Towards a systematic screening tool for quality assurance and semiautomatic fraud detection for images in the life sciences. Sci Eng Ethics. 2017;23:1113–28.